Author's response to reviews

Title: Myasthenia gravis and pregnancy: Clinical implications and neonatal Outcome

Authors:

Jose Francisco Tellez-Zenteno (jftellez@quetzal.innsz.mx)
Lizbeth Hernandez-Ronquillo (liztellez33@hotmail.com)
Vicente Salinas (vicente_salinas@latinmail.com)
Guillermo Garcia Ramos (garcira@quetzal.innsz.mx)
Bruno Estanol (bestanol@hotmail.com)
da Silva Orlando (odasilva@julian.uwo.ca)

Version: 6 Date: 7 July 2004

Author’s response to reviews:

Thank you for the commentaries.

Responses

Main observation

We have corrected the mistake. We have 17 deliveries; 8 were by C-section, 8 were vaginal deliveries and 1 was vaginal with forceps. Finally we have only 17 seventeen infants. The first version had a mistake in the number of newborns.

Minor observations

1.-We have included the numbers of the newborns. They are included in the results and in table 2.

The information is the following;

Seventeen infants were born at mean +/- SD gestational age of 37.5+/-3.0 weeks and a mean birth weight of 2710 +/- 73 g. Only one infant presented with transient neonatal myasthenia gravis. This baby presented with sucking difficulties, which resolved spontaneously by day 7 and did not require any specific treatment. The patient with myasthenic crisis delivered by spontaneous vaginal delivery at 37 weeks. The weight of the newborn was 2800 g without complications. No congenital anomalies were identified in any of the newborns.

2.-The article has been reviewed for an expert in the English language to correct the wrong use of English.

3.-We deleted table 3; the information was included in the discussion. We have corrected the missing values in tables and the wrong use of terms.